Author's response to reviews

Title: Elastic Stable Intramedullary Nailing (ESIN), Orthoss(R) and Gravitational Platelet Separation- System (GPS(R)): An effective method of treatment for pathologic fractures of bone cysts in children

Authors:

Marion Rapp (rapp@paedia.ukl.mu-luebeck.de)
Daniel Svoboda (daniel.svoboda@umm.de)
Lucas M Wessel (lucas.wessel@umm.de)
Martin M Kaiser (kaiser@medinf.mu-luebeck.de)

Version: 2 Date: 1 December 2010

Author's response to reviews:

Dear Editor

We add a paragraph about ethics in our method section and included an Authors' contributions section before the Acknowledgements and Reference list.

Acknowledgements: There are no further person acknowledge who contributed towards the study and no funding occured.

Sincerely yours
Marion Rapp